

Received 8 September 2023, accepted 21 September 2023, date of publication 28 September 2023, date of current version 9 October 2023.

Digital Object Identifier 10.1109/ACCESS.2023.3320570



Automatic Segmentation of Pancreas and Pancreatic Tumor: A Review of a **Decade of Research**

HIMALI GHORPADE¹, JAYANT JAGTAP^{®2}, (Senior Member, IEEE), SHRUTI PATIL⁰³, (Member, IEEE), KETAN KOTECHA^{03,4}, AJITH ABRAHAM^{65,6}, (Senior Member, IEEE), NATALLY HORVAT⁶⁷, AND JAYASREE CHAKRABORTY⁰⁸, (Senior Member, IEEE)

¹Symbiosis Institute of Technology, Symbiosis International (Deemed University), Pune 412115, India

²Artificial Intelligence and Machine Learning Department, Symbiosis Institute of Technology, Symbiosis International (Deemed University), Pune 412115, India

³Symbiosis Centre for Applied Artificial Intelligence, Symbiosis Institute of Technology, Symbiosis International (Deemed University), Pune 412115, India

⁴Department of Computer Science, UCSI University, Kuala Lumpur 56000, Malaysia

⁵School of Computer Science Engineering and Technology, Bennett University, Greater Noida, Uttar Pradesh 201310, India

⁶Center for Artificial Intelligence, Innopolis University, Innopolis, 420500 Tatarstan, Russia

⁷Department of Radiology, Memorial Sloan Kettering Cancer Center, New York, NY 10065, USA ⁸Department of Surgery, Memorial Sloan Kettering Cancer Center, New York, NY 10065, USA

Corresponding author: Javant Jagtap (javant.jagtap@sitpune.edu.in)

This work was supported by the Analytical Center for the Government of the Russian Federation under Agreement 70-2021-00143 dd.01.11.2021 and Agreement 000000D730321P5Q0002 (IGK).

ABSTRACT In the current era of machine learning and radiomics, one of the challenges is the automatic segmentation of organs and tumors. Tumor detection is mostly based on a radiologist's manual reading, which necessitates a high level of professional abilities and clinical experience. Moreover, increasing the high volume of images makes radiologists' assessments more challenging. Artificial intelligence (AI) can assist clinicians in diagnosing cancer at an early stage by providing a solution for assisted medical image analysis. The automated segmentation of tumor is better realized through conventional segmentation methods and, nowadays, through machine learning and deep learning techniques. The segmentation of abdominal organs and tumors from various imaging modalities has gained much attention in recent years. Among these, pancreas and pancreatic tumor are the most challenging to segment and have recently drawn a lot of attraction. The main objective of this paper is to give a summary of different automated approaches for the segmentation of pancreas and pancreatic tumors and to perform a comparative analysis using various indices such as dice similarity coefficient (DSC), sensitivity (SI), specificity (SP), precision (Pr), recall and Jaccard index (JI), etc. Finally, the limitations and future research perspectives of pancreas and tumor segmentation are summarized.

INDEX TERMS Deep learning, machine learning, pancreas segmentation, pancreatic ductal adenocarcinoma, tumor.

I. INTRODUCTION

In clinical practice, radiologists help in the visual analysis of various anatomical structures. Small changes in form, size, or structure can indicate illness and aid in the confirmation

The associate editor coordinating the review of this manuscript and approving it for publication was Yiming Tang $^{\square}$.

of a diagnosis. Manual readings with radiographic images, such as computer tomography scan (CT) or magnetic resonance imaging (MRI), are tedious and can lead to inter and intra-operator variability. Moreover, for quantitative radiographic image analysis with machine learning which has shown widespread application in clinical decision making, segmentation of organs or tumor plays an important role.

© 2023 The Authors. This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/



FIGURE 1. Major challenges in segmentation of pancreas and pancreatic tumor.

Automatic systems with artificial intelligence can assist radiological experts in detecting and diagnosing disease and thus improve treatment management. One such disease that requires automated segmentation is pancreatic cancer. One of the most serious illnesses, pancreatic cancer, is becoming more common. According to the global cancer observatory (GLOBOCAN) 2020 statistics, pancreatic cancer accounted for approximately 466,003 deaths worldwide, with 54,277 fatalities reported in the United States in the same year [1].

The most prevalent kind of pancreatic cancer, pancreatic ductal adenocarcinoma (PDAC), arises from the exocrine glands and ducts of the pancreas [2]. Despite improvement in treatment techniques for cancer care, five year survival rate for PDAC is only 10% [3] due to its late diagnosis and lack of effective treatment. More than half of the patients are with metastasis, and 30% with locally advanced disease at the time of diagnosis. As the mortality and incidence rate of pancreatic cancer is continually rising globally, there is an unmet need to enhance the survival outcomes of individuals affected by this disease through the implementation of advanced diagnostic and therapeutic interventions. Recent studies show that patients diagnosed at stage-I can have the most favorable outcome, with a 5-year survival rate reaching up to 80% [4]. Thus, better detection of early-stage disease is a tremendous opportunity to improve PDAC prognosis.

However, detection and segmentation of PDAC is often challenging and vary due to irregular contours and ill-defined margins [5], as shown in FIGURE 1. In addition, in the past decade, with the advancement of imaging technology, radiographic images are being widely investigated with machine learning to develop imaging biomarkers of diagnosis, progression, outcome, and response prediction [6]. However, these techniques highly depend on manual segmentation. As compared to the liver, spleen, and other abdominal organs, the segmentation of the pancreas is challenging as pancreas shape, size, and position are different between individuals [7], [8], [9].

Automated segmentation of pancreas and pancreatic tumor techniques thus can help radiologists not only in proper detection and diagnosis but also in developing more generalized imaging biomarkers for pancreatic cancer. Several approaches have been proposed for automated segmentation of pancreas and pancreatic tumor using different techniques. However, the methods with unsupervised learning, such as clustering, region growing, threshold based methods, etc., did not provide satisfactory performance.

Deep learning-based segmentation techniques have recently seen widespread implementation and have outperformed traditional segmentation techniques in terms of performance. These models consist of hierarchical architecture with different layers. Deep learning with convolutional neural networks (CNN) is the most successful architecture for image analysis. Neural networks, consisting of neurons with parameters and activation functions, have been utilized to extract and combine image features, enabling the development of diagnostic models.

The neural network is composed of neurons with parameters and activation functions to extricate and merge the image features, enabling the development of diagnostic models. In diagnosis and segmentation of various diseases such as diabetic retinopathy [10], liver masses [11], and skin cancer [12] CNN has achieved a better accuracy than conventional methods. Discovering the practicality of CNN in pancreatic cancer segmentation has major implications discussed in pertinent sections. This paper aims to comprehensively review the studies based on the segmentation of the pancreas and pancreatic tumor by using various conventional, unsupervised and supervised approaches, including deep learning methods. The paper also discusses the current challenges in pancreatic tumor segmentation and the future scope of such techniques.

FIGURE 2 represents the detailed outline of the literature review paper.

The subsequent sections of the paper are as follows. Section II focuses on the database selection method. Section III describes the statistical analysis of AI in pancreatic cancer segmentation. Section IV addresses various imaging modalities. Section V presents a detailed review of various pancreas and pancreatic tumor segmentation techniques. Section VI describes the evaluation metrics used in segmentation. Section VII provides an overview of the experimental datasets. Section VIII discusses the findings and insights derived from the review. Section IX summarizes the overall review. Finally, Section X concludes the paper by outlining the future approaches for pancreas and tumor segmentation.

II. DATABASE SELECTION METHOD

A comprehensive search for pancreas and pancreatic tumor segmentation was performed to identify relevant scholarly articles for this survey. The search included prestigious scientific papers from reputed publishers in digital libraries like ScienceDirect, Springer, IEEE Xplore, PubMed, etc. Additionally, annual challenges like the Medical Segmentation Decathlon (MSD) were reviewed. In addition, the

IEEE Access



FIGURE 2. Outline of Literature Review.

search extended to include databases such as Google Patents, and the Web of Science, where search keys such as "pancreas", "pancreatic cancer", "segmentation", "artificial intelligence", "machine learning", and "deep learning" were used.

The title and abstract of the articles were assessed as a part of the screening process, and then the full text review was performed for the selected articles. The collected data included information about the authors, article/report title, year, imaging modality, segmentation method, dataset, and algorithm performance. The goal of this survey is to offer a comprehensive analysis of techniques employed for pancreas



FIGURE 3. Implementation of literature review.



FIGURE 4. Analysis of top 15 countries with publication records.

and pancreatic tumor segmentation. FIGURE 3 represents flow of literature review conducted.

III. STATISTICAL ANALYSIS OF PANCREATIC CANCER SEGMENTATION

Web of Science database is used for the analysis of pancreatic cancer segmentation. Web of Science is one of the most common and acknowledged databases in bibliometric analysis. The search terms used were "pancreatic" OR "pancreas" OR "image segmentation" OR "artificial intelligence" OR "machine learning" OR "deep learning" OR "convolutional neural network". The publication date range was restricted to 2013 to 2023. The documents were restricted to articles and reviews. VOSviewer and Microsoft Excel are used for the representation of analysis.

A. QUANTITATIVE ANALYSIS

The quantitative analysis provides an overview of various countries having research in artificial intelligence for pancreatic cancer and an analysis of total publications and citation records for pancreatic cancer segmentation across the globe.

Major countries around the world have received attention for research in pancreatic cancer segmentation using AI. Among those, 15 countries with the highest publication record for pancreatic cancer were observed and shown in FIGURE 4. According to the figure, China is the country with the highest number of publication records for pancreatic cancer, followed by the United States.

The analysis of the total number of publications worldwide and the total number of citations from 2013 to 2023 is depicted in FIGURE 5. The graph clearly illustrates a



FIGURE 5. Number of publications.



FIGURE 6. Co-occurrence of keywords for pancreatic cancer segmentation by VOSviewer.

consistent upward trend in the number of publications focusing on pancreatic cancer segmentation utilizing artificial intelligence.

B. QUALITATIVE ANALYSIS

Qualitative Analysis includes the co-occurrence analysis that identifies hot topics and aids scholars in better understanding current scientific issues by examining the keywords in a group of publications. In this review, VOSviewer software was used to analyse the co-occurrence of keywords from the Web of Science database. Qualitative analysis was done on both pancreas and pancreatic cancer segmentation. FIGURE 6 represents a visual network map of the co-occurrence of keywords for pancreatic cancer, and FIGURE 7 shows a network map of the co-occurrence of keywords for the pancreas. The visualization nodes are depicted in various colors to represent different clusters. The node size corresponds to the occurrence of keywords associated with it. The greater the size of the node higher the occurrence of keywords. Additionally, a thick connecting line between the nodes indicates a strong relation between the items, highlighting their close association.

IV. IMAGING MODALITIES

Pancreatic cancer diagnosis and evaluation heavily rely on different imaging modalities, each offering unique



FIGURE 7. Co-occurrence of keywords for pancreas segmentation by VOSviewer.

Types of Modality for Detection, Diagnosis & Segmentation	Advantages of Imaging Modality	Disadvantages of Imaging modality
EUS scan	 Provides good resolution for smaller lesions. Provides real time imaging. Safe and minimal invasiveness. 	 Dependent on expertise. Limited field of view.
CT scan	 CT scans are widely available. Provides a broader view for pancreatic cancer segmentation. 	 Exposes patients to ionizing radiation. Uses iodine contrast agents which can cause allergies to patients.
MRI scan	 Excellent soft tissue contrast. Doesn't use ionizing radiation 	 Longer duration of scanning. Higher sensitivity to motion artifacts.
PET scan	 Provides imaging of entire body which helps in detecting distant metastases 	 Lower spatial resolution. May produce false positive results in case of inflammation.

FIGURE 8. Imaging modalities.

advantages and capabilities. The common imaging modalities used for the detection, diagnosis, and prognosis of pancreatic cancer are CT, MRI, Positron Emission Tomography (PET) contrast, and enhanced-endoscopy ultrasound (CE-EUS). For the assessment of the pancreas and pancreatic cancer, computed tomography (CT) is frequently used as the primary imaging modality. It provides extensive anatomical information, identifying pancreatic lesions and surrounding structures by exploiting variations in tissue density. The ability to discriminate between healthy pancreatic tissue and tumors based on their vascular characteristics is further improved by enhanced CT. Using CT imaging, the perivascular vascularity around the pancreas may be seen clearly.

Magnetic Resonance Imaging (MRI) is another important modality to observe the pancreas and other organs. MRI scans can help to visualize detailed and in depth properties of organ scans similar to CT. Different imaging sequences can be used by MRI to pinpoint the location of lesions and identify minute anomalies. It is vital to note that CT imaging often provides abdominal imaging with a better degree of precision than MRI.

Another specialized imaging technique, Contrastenhanced Endoscopy Ultrasound (CE-EUS), is also used as an effective visualization method for pancreas and lesions. In this imaging technique, high-resolution images are taken using an endoscope with an ultrasonic probe that is inserted into the digestive tract. Accurate identification and characterisation of pancreatic tumors are made possible by real-time imaging with good contrast provided by CE-EUS.

Positron Emission Tomography (PET) is another imaging modality that can also be used for pancreatic tumor visualization. A radiotracer that builds up in places with high metabolic activity, such as cancer cells, is injected in this procedure. PET scans are able to locate and identify pancreas and pancreatic tumors, as well as determine the size and likelihood of metastases.

In pancreas and pancreatic cancer evaluation, each modality has its own strengths and weaknesses. There are various factors for the choice of the modality, such as clinical scenario, availability, cost, and expertise. In this literature, the focus will be more on the CT scan modality. FIGURE 8 illustrates different imaging modalities with advantages and disadvantages for pancreas and tumor detection, diagnosis, and segmentation.

V. PANCREAS AND TUMOR SEGMENTATION

The process of medical image segmentation involves utilizing computer-based image processing techniques to extract region of interest from medical images, such as to accurately identify and isolate specific organs, tissues, and tumors from CT and MRI for detection, diagnosis, or for further studies. Segmentation basically divides the image into different sub-regions based on the similarity or differences between regions. Medical image segmentation enables clinicians to examine lesions and other region of interest (ROI) both qualitatively and quantitatively, which can improve the precision of medical diagnosis.

Different automated pancreas and pancreatic tumor segmentation approaches are explained in the literature. The utilization of automated segmentation techniques can alleviate the difficulties associated with manually assessing pancreatic tumors. In this section, first, different pre-processing techniques will be discussed. Next conventional segmentation techniques followed by supervised machine learning and deep learning methods for pancreas segmentation and finally the paper delves into the methods for pancreatic tumor segmentation with their reported performance.

A. PRE-PROCESSING TECHNIQUES TO REMOVE NOISE AND ENHANCE THE IMAGE QUALITY

Pre-processing medical images is crucial before performing any image analysis or segmentation. Pre-processing improves the image quality and helps remove the noise for better outcomes in terms of accuracy and reliability for segmentation tasks. In medical images, noise occurs during

tasks. In medical

acquisition, specifically when there is a low radiation dose. Handling different types of noise is necessary for accurate and precise segmentation of pancreas and pancreatic tumors. Some of the most common noises are salt and pepper noise, Gaussian noise, speckle noise, motion artifacts, etc. Different filtering techniques are incorporated to reduce this noise. Previously, the noise was removed by traditional filtering techniques such as mean, median, Gaussian, and Weiner filters. Nowadays, deep learning based methods such as deep CNN [12], CNN denoising autoencoder [13], Generative Adversarial Networks (GAN) [15], [16], conditional GAN [17], [18], etc. are used to remove noise from medical images. Other pre-processing operations, including intensity normalization, cropping, resizing, and data augmentation techniques, are performed to improve the quality of CT scans. Intensity normalization is standardizing the pixel values such that images should have consistent pixel values for segmentation. Cropping focuses on the region of interest and discards the area outside that region. Resizing helps to change the resolution or spatial dimensions of the CT scans to achieve the desired resolution. Data augmentation helps to increase the training data artificially. Different augmentation techniques, such as rotation, scaling, zooming, and flipping, are applied to improve the robustness of machine learning models for limited training data. The choice of pre-processing should align with the medical imaging modality and segmentation task. Pre-processing plays a vital role in enhancing the quality and effectiveness of medical images. Pre-processing also helps to mitigate issues such as acquisition settings and noise. It is essential to consider pre-processing as an integral component of the medical image segmentation pipeline to achieve good results.

B. AUTOMATIC METHODS FOR PANCREAS SEGMENTATION

1) CONVENTIONAL SEGMENTATION METHODS

In conventional segmentation techniques, the multi-organ atlas-based method is among the common approaches for medical image segmentation. The "multi-organ" name indicates that this method segments multiple organs, and the name "atlas" indicates a labeled reference image or predefined template of an anatomical structure. The atlas can be the pancreas, liver, spleen, and any other abdominal organ. The atlas is obtained from CT images wherein the organs are manually segmented and labeled, which is further used as a reference for implementing different segmentation algorithms. A pre-existing atlas or template of the pancreas can be used further to guide the segmentation process.

During the segmentation stage, the atlas or template is matched with the target CT scan using image registration techniques. The labeled information is transferred to the associated structures, such as the pancreas in the target image, by registering the atlas with the target image.

The segmentation algorithm can precisely detect and outline the pancreas in the target CT image with the help



FIGURE 9. Atlas-based segmentation of pancreatic tumor.

of an atlas which provides valuable insights about the shape and location of the pancreas. The algorithm may overcome difficulties seen in CT scans, such as blurred borders and low contrast, by making use of the atlas.

Overall, the use of an atlas in CT image analysis makes it easier to segment anatomical structures automatically by giving a reference template and prior knowledge of the predicted appearance and spatial connections of organs, such as the pancreas. A variety of therapeutic applications, such as the identification and planning of pancreatic cancer, are made possible by this method's improved precision and efficacy in pancreas segmentation in CT scans. The multiorgan atlas-based segmentation method offers the benefit of low training time. The atlas-based segmentation approach is represented in FIGURE 9.

Shen et al. [13] proposed an automated segmentation method for various organs in the abdomen and adipose tissue compartments. Their approach utilized a multi-atlas registration technique, involving the manual creation of 19 atlases, followed by registration-based segmentation. The method was evaluated on data from 26 obese patients pre- and post-weight-loss intervention, demonstrating good agreement with manual segmentation.

In a separate study by Tong et al. [14], a technique for multiple organ segmentation using CT scans was developed. They employed two DDLS (discrimination-based dictionary learning) methods. In the first, called global DDLS (G-DDLS), a set of atlases was chosen depending on how closely they resembled the target image within the global mask. The second, local DDLS (L-DLS), used voxel-wise atlas selection to choose similar atlases locally at various locations inside the target image.

Wolz et al. [15] developed a fully automated multi-organ segmentation that involves a hierarchical atlas generation step and a refinement step. The atlas labels generated during this step are used to describe and define anatomical structures at different scales - global (entire image), organ (individual organ), and voxel level(pixels). The most suitable atlases are selected based on the global image appearance, aligned with the target image, and weighted locally on an individual organ. Subsequently, a patch-based segmentation refinement is performed at the voxel level. To enhance the segmentation, a graph-cuts based refinement step is conducted, incorporating constraints related to local smoothness and high-level spatial relationships.

In order to choose atlases with a high degree of pancreatic similarity to the unlabeled volume, Karasawa et al. [16] implemented a structure specific atlas generation that used structural information in the generation of the atlas. Chu et al. [17] generated probabilistic atlases to segment the pancreas and other abdominal organs using maximum a posterior (MAP) estimation and a graph cut method. Saito et al. [18] use a statistical shape model to consider all possible shapes and a search algorithm to select the best shape for pancreas segmentation. This approach does not require predefined shapes or complex hierarchies. The algorithm optimizes both the shape model and the segmentation labeling, resulting in more precise and efficient segmentation. A regression forest technique was implemented by Oda et al. [19] to evaluate the size of the pancreas, position of the pancreas, and a patient specific atlas generation wherein a new similar atlas was generated based on information related to blood vessel characteristics. The segmentation process employed a combination of the expectation maximization (EM) algorithm, utilizing atlases as priors, and the graph-cut optimization method. The utilization of the multi-organ atlas-based method can significantly enhance the results of pancreas segmentation, playing a vital role in applications like diagnosis and pancreatic cancer treatment.

2) SUPERVISED MACHINE LEARNING & DEEP LEARNING BASED SEGMENTATION OF PANCREAS

Machine learning based techniques are classified into two major techniques supervised and unsupervised learning. Supervised machine learning based pancreas segmentation techniques have recently come to light as potential alternatives to the challenges faced by the traditional segmentation approach. These methods use huge of annotated medical image data to train the models that accurately identify and segment the pancreas. Figure 10 shows the evolution of pancreatic cancer segmentation from the 1950s using machine learning and deep learning.

Deep learning uses multiple layer artificial neural networks (hence "deep") to learn complex patterns and representations directly from the data. The training of the neural network involves a vast collection of labeled medical images, where the ground truth segmentation of the target structures is provided for training purposes. The network learns to autonomously extract pertinent features and patterns from the input image and maps them to their respective segmentation masks. Convolutional neural networks (CNNs) play a pivotal role in deep learning by directly extracting features from the images. Recently deep learning techniques can achieve state-of-the-art results in pancreas segmentation tasks. These methods have the potential to improve the accuracy and



FIGURE 10. Evolution of machine learning and deep learning for pancreatic cancer segmentation.

efficiency of pancreas segmentation, which can lead to better diagnosis, treatment, and patient outcomes. Convolutional neural networks (CNNs) are widely employed to segment the pancreas, due to their effective learning characteristics that allow them to directly learn from imaging data. Several CNN-based techniques like deep CNNs [20], VoxResNet [21], SegNet [22], fully convolutional neural networks (FCNs) [23], and U-Net [24] have shown promising results in performing pixel-level labeling tasks for semantic segmentation. These algorithms employ different architectures and techniques such as skip connections, pooling, and upsampling, to increase the segmentation accuracy and efficiency. For instance, the U-Net employs a contracting path and an expanding path for feature extraction and localization. The VoxResNet architecture combines residual learning with a 3D CNN to capture both spatial and temporal information.

Zhou et al. [25] proposed a "coarse to fine" model for pancreas segmentation that involves the use of a coarse-scaled and fine-scaled network. The framework is built upon the observation that processing smaller input regions can achieve more precise segmentation outcomes. Despite being trained and tested independently, this approach has been shown to deliver better accuracy.

Additionally, they showcased a saliency transformation network [26], which includes the generation of spatial weights using the score map obtained from the coarse-scaled segmentation network and applying these weights to the fine-scaled segmentation network. This method allows both segmentation networks to be adjusted together, making them more efficient.

The holistically nested network, initially developed for edge detection using deep learning, was repurposed for image segmentation and showed promising results in pancreatic segmentation [27], [28]. It uses deep dense per pixel masking to process different sequences of 2D image slices. However, it did not explore explicitly applying a spatial consistency requirement on slice segmentation. Zhu et al. [29] implemented a ResNet architecture on 3D data for segmentation of the pancreas. This framework takes advantage of extensive spatial information and all three dimensions of data, leading to better segmentation results than the 2D counterpart.

Roth et al. [30] implemented a cascaded dual stage architecture that consists of pancreas localization and segmentation. In the first stage, the pancreas is localized, and a robust bounding box is produced for more comprehensive segmentation in the second stage. To segment the pancreatic tissue, a holistically nested convolutional network (HCNN) is used, which takes into account three orthogonal views: coronal, axial, and sagittal. The HCNN combines multiple convolutional layers to capture multi-scale features, which leads to improved segmentation results. Pooling is used to concatenate the HCNN probability maps for every pixel, which helps to generate a 3D bounding box of the pancreas and improve recall. This approach is advantageous since it reduces the likelihood of missing any part of the pancreas during segmentation, which can be critical for the detection of pancreatic diseases.

Attention U-Net was proposed by Oktay et al. [31] for pancreas segmentation. The model combines the attention mechanism and U-Net architecture to selectively highlight regions of the input image that are relevant for pancreas segmentation. This attention mechanism is learned from the input image itself, which allows to focus on informative regions and suppress irrelevant regions. The Attention U-Net comprises an encoder and decoder layer with skip connections between them. The attention mechanism is added in the decoder part of the network, where it performs element-wise multiplication of the generated spatial attention maps with the feature maps from the encoder layers. These attention maps are learned by the model during training to highlight the relevant features for pancreas segmentation. On two distinct datasets, the model was assessed and achieved an exceptional outcome. A detailed analysis of the attention



FIGURE 11. History of different pancreas and pancreatic tumor segmentation methods.

maps was performed, showing that the model indeed learns to focus on the pancreas region and suppress irrelevant background regions.

For pancreas segmentation, Man et al. [32] introduced a novel approach based on deep reinforcement learning in CT images using a geometry-aware deformable U-Net. Their approach involves a two-stage process, with the first stage focusing on, a single slice of the 3D CT volume is selected, and a reinforcement learning agent is trained to identify the optimal bounding box for the pancreas in that slice. In the second stage, the identified pancreas region is segmented using the deformable U-Net, which utilizes the shape information of the pancreas in different slices. The pixel-wise results from the three orthogonal axes (coronal, axial, and sagittal) are combined to generate the final segmentation. Their approach demonstrated superior performance compared to other existing methods for pancreas segmentation.

Xue et al. [33] developed a method of two cascaded phases, where the first phase localizes the pancreatic region using a shape-specific module that extracts shape-specific features from the CT image. For precise pancreas segmentation, the second phase uses a multitask 3D dense-U-Net. The proposed method surpasses various advanced techniques, such as U-Net, Attention U-Net, and Shape-aware U-Net, in terms of segmentation performance and handles inter-slice inconsistencies using a slice-to-slice fusion mechanism. The suggested strategy is effective and accurate for the clinical use of pancreas segmentation.

Farag et al. [34] introduced a novel bottom-up approach for segmenting the pancreas consisting of two main stages. In the first stage, the authors proposed a superpixel-based segmentation method to generate an initial segmentation mask, which is then refined using a cascaded framework. In the second stage, the authors proposed a deep image patch

108734

labeling method to classify each image patch as pancreas or non-pancreas using a deep CNN.

Chen et al. [35] presented a method that incorporates a multi-scale supervision approach and two-view feature learning for pancreas segmentation using CT images. The method effectively captures the multi-view complementary information of the pancreas and combines features at different scales to improve the segmentation accuracy.

A MobileNet-U-Net (MBU-Net) is proposed by fusing MobileNet-V2 and U-Net architecture with repetitive dilated convolutions for semantic pancreas segmentation [36].

Qiu et al. [37] introduced a novel framework called dual enhancement module, that enhances fine-scale segmentation input from the coarse-scaled segmentation mask. Segmentation performance was improved using multi-scale feature calibration U-Net (MFCUNet) architecture at the pixel level, as directly fusing these features for recovering boundary information can lead to redundancy and inaccuracies. Additionally, a cascaded MFCUNet was implemented, that combines the merits of both MFCUNet and dual enhancement module, achieving the best pancreatic segmentation performance possible. Overall, the proposed framework demonstrates significant improvements in pancreatic segmentation using medical images. He et al. [38] developed a model combining U-Net with a transformer called U-Netmer for the segmentation of medical images. The U-Netmer model offers flexibility in segmenting input images with various patch sizes while maintaining the same structure and parameters. This unique design and clever training strategies enable the U-Netmer to effectively integrate multi-scale contextual knowledge during the learning process.

Wang et al. [39] implemented a two input approach for pancreas segmentation, which includes a graph based visual saliency (GBVS) algorithm and a v-mesh FCN which helps to enhance feature extraction and reduces the semantic gap, an attention mechanism combines multiple feature maps to emphasize ROIs.

Li et al. [40] implemented a probabilistic-map-guided bidirectional recurrent UNet (PBR-UNet) for the segmentation of the pancreas. It combines intra-slice information and inter-slice probabilistic maps to create a local 3D hybrid regularization strategy, and a bi-directional recurrent optimization scheme is followed to improve the accuracy.

Quereshi et al. [41] implemented a multiphase deep learning framework for precise pancreatic segmentation in CT imaging. The pancreas is localised using the VGG-19 deep learning network, and then soft labels are created in the localised area. A 3D volume template that depicts the overall form of the pancreas is then combined with the soft labels. This fusion process allows for the refinement of soft labels and leads to improved segmentation results.

Giddwani et al. [42] proposed a multi-rate deep dilated V-Net architecture, which demonstrated better segmentation performance. Mo et al. [43] developed a 3D iterative enhancement network that accurately segmented the pancreas. They used a residual network for feature extraction and refined individual features. Nishio et al. [44] used standard and deep U-Nets for the segmentation of the pancreas. Zhu et al. [45] integrated a neural architecture search to find the optimal architecture between 2D, 3D, or pseudo 3D convolution at each layer. They implemented neural architecture search on NIH and MSD dataset.

Fan et al. [46] implemented a regularized U-Net architecture to achieve regularized pancreas segmentation. Similarly, other U-Net architectures, such as fully-convolutional U-Net [47], attention U-Net [31], and automatic multi-organ segmentation with adversarial loss [48], have been developed to achieve better segmentation results. Dai et al. [49] developed a Trans-Deformer network using a combination of 2D U-Net at the coarse stage and added a deformable convolution to the vision transformer at the fine stage. They implemented the architecture on both the NIH and MSD datasets. Thus application of supervised machine learning and deep learning based algorithms is widely implemented for the segmentation of the pancreas.

3) UNSUPERVISED MACHINE LEARNING BASED SEGMENTATION

Unsupervised machine learning is the algorithm that learns patterns and structures from unlabeled data without explicit guidance or supervision. Examples of unsupervised learning methods consist of K-means clustering [50], Principal Component Analysis (PCA) [51], and Non-negative Matrix Factorization (NMF) [52]. Very few papers represent the segmentation of the pancreas and tumor using unsupervised learning. Roy et al. [53] used an improved K-means clustering for tumor segmentation. Jain et al. [54] developed an unsupervised approach that localizes the pancreas from CT scans. Clustering-based methods evaluate the clustering quality using the Clustering Validity Index (CVI). The

VOLUME 11, 2023

clustering index helps researchers to assess and compare different clustering algorithms. There are different types of CVIs, such as the Davies-Bouldin Index, Silhouette Score, Dunn Index, etc. Tang et al. [55] developed a novel CVI called the Triple Center Relation (TCR) index for fuzzy clustering. It considers two factors, within-class compactness and between-class separateness, which is a vital property of the TCR index. This index is robust and can achieve good results for high-dimensional datasets. The application of unsupervised learning techniques is uncommon in pancreas and pancreatic tumor segmentation research.

C. DEEP LEARNING BASED SEGMENTATION OF PANCREATIC TUMOR

Segmentation of pancreatic tumor and cyst is quite challenging as compared to segmentation on the pancreas. Deep learning has become a popular technique for pancreatic tumor segmentation due to its ability to automatically extract intricate patterns and features from extensive datasets. Figure 11 shows the overall history of different pancreas and pancreatic tumor segmentation methods. A detailed illustration of machine learning and deep learning based segmentation is represented in Figure 12. This section presents a review of various methods used for pancreatic tumor segmentation.

Du et al. [56] implemented a novel multi-scale channel attention UNet architecture to segment pancreatic tumor. The multi-scale network was embedded into the encoder and decoder for semantic information extraction. This work was implemented on the private dataset and focused on segmenting small pancreatic tumors.

Li et al. [57] performed the segmentation with a 3D FCN. To enhance tumor segmentation without compromising the pancreatic information, they employed three different guided modules for temperature. The balancing temperature loss function was created with the specific purpose of dynamically adjusting the learning points between the pancreas and the tumor, ensuring a balanced selection of features. Additionally, they implemented a rigid temperature optimizer to probabilistically accept non-improving movements and adaptively avoid local optima. Furthermore, they incorporated a soft temperature indicator that automatically guided the network towards a fine-tuning phase as the model reached stability. This approach led to improved segmentation outcomes, ensuring both accuracy and efficiency in jointly segmenting the pancreas and pancreatic tumors. Overall, their methodology showcased promising results in this regard.

Zhu et al. [58] introduced a framework that combines segmentation and classification for interpreting abnormality in medical images. The framework involves training a segmentation network to identify tumor voxels and then performing classification based on the presence of these voxels in testing volumes. Two important techniques were employed to enhance classification accuracy: multi-scale network training and coarse-to-fine testing. Significantly,



FIGURE 12. Illustration of machine learning and deep learning-based segmentation techniques.

these strategies helped the classification process perform better overall.

Chen et al. [59] developed a novel approach unified tumor Transformer (UniT) model for the simultaneous diagnosis, detection, and segmentation of eight common cancers from 3D CT scans. UniT leverages a query-based transformer architecture and introduces a clinically inspired hierarchical tumor representation. It incorporates a dual-task query decoding stage to generate segmentation masks. Oh et al. [60] introduced a method for the segmentation of pancreatic cystic lesions from endoscopic ultrasonography (EUS) images. They developed an attention U-Net architecture, which incorporates an attention mechanism that enables concentrating on the informative regions of input images. This architecture was assessed on both internal and external test datasets and a comparison with other cutting-edge models like Basic U-Net, Residual U-Net, and U-Net++.

Alves et al. [61] developed a nn-UNet for the detection and segmentation of pancreas and pancreatic tumor. This architecture is an extension of the 3D U-Net model which also includes a deep supervision mechanism to enable better training of the network. The nn-UNet architecture achieved better performance for the pancreas and showcased better results for tumor detection. This architecture is capable of simultaneously segmenting multiple anatomical structures.

Si et al. [62] segmented pancreatic tumor using a combination of three different architectures. The pancreas was detected using ResNet-18, and U-Net32 was used for the segmentation of the pancreas region from the

CT scan. Finally, ResNet-34 was implemented for tumor detection. This approach demonstrated better accuracy for both pancreas and tumor detection.

Mahmoudi et al. [63] developed a framework for pancreas and tumor segmentation by combining the CNN architecture with textured U-Net architecture. Initially, the pancreas was localized by CNN and then segmentation of tumor was performed using textured U-Net architecture. The framework showed better results for the pancreas than the pancreatic tumor segmentation.

Iwasa et al. [64] presented an automated system for the segmentation of PDAC from contrast-enhanced endoscopic ultrasound (CE-EUS) video images. The system utilized a U-Net and performed training and evaluation using a 4-fold cross-validation. This method showed good results in segmenting PDAC from CE-EUS images and has the potential to aid in the early diagnosis and treatment of PDAC. Li et al. [65] introduced an enhanced UNet framework called Position Guided Deformable UNet (PGDUNet) architecture. With regard to tumor segmentation, this architecture handles issues including size and form variations and significant class imbalance. PGDUNet consists of a deformable convolution with a localization route and a focal loss function for the suppression of noise. The study specifically focuses on both the pancreas and pancreatic tumor segmentation.

Tureckova et al. [66] developed a novel approach to improve the accuracy of pancreatic tumor segmentation. Deep supervision and attentional gates were used in their network architecture to enhance the segmentation accuracy.



FIGURE 13. Example of abdominal CT scan of the pancreas and PDAC segmentation. (A) A non-segmented CT scan of the abdomen (B) Segmented pancreas in blue, tumor in red, and duct in green colour [69].

The use of attentional gates helped to focus on the vital regions, while deep supervision helped in better training of the network.

Jiang et al. [67] implemented DLU-Net for the segmentation of pancreatic tumor edges. The model is proposed of a densely connected U-Net architecture combined with an attention gate module and deep supervision mechanism. The proposed method can assist clinicians in more accurate tumor segmentation and treatment planning.

Liang et al. [68] developed an automated system for pancreatic tumor segmentation using multi-parametric MRI scans. The system consists of three different modules: feature extraction, classification, and segmentation. This system was implemented on a private dataset giving better results for the segmentation of tumor. The sample abdominal CT scan of the pancreas and pancreatic tumor segmentation is represented in FIGURE 13.

Zheng et al. [70] developed a framework for the segmentation of the pancreas using a squeeze and extraction block in U-Net architecture. The uncertain regions were also determined using shadowed sets. The framework was implemented using MRI as well as CT scans. The MRI dataset was used to segment the tumor and the CT scan dataset segmented the pancreas. This makes the framework versatile in clinical practice.

Guo et al. [71] segmented pancreatic tumor using combination of U-Net and Layered Optimal Graph Image Segmentation for Multiple Objects and Surfaces (LOGIS-MOS) approach. The LOGISMOS approach incorporates the geometric and spatial information in the segmentation. The graph-based approach of LOGISMOS helps user to adjust the segmentation by changing nodes and edges. This method enables to precisely segment pancreatic tumor.

Zhou et al. [72] implemented Deep FCN for segmentation of pancreas and cyst. This method was implemented on private dataset of 131 CT scans. Deep FCN approach demonstrated better results for pancreas segmentation and quite good results for segmentation of cyst.

Thus precise and reliable segmentation of pancreatic tumor is achieved using deep learning techniques. Table 3 summarizes the various deep learning based methods and its performance for pancreatic tumor and cyst segmentation. The majority methods have been evaluated on publicly available as well as privately available datasets, demonstrating their potential for clinical applications. Furthermore, the advancement of deep learning-based segmentation techniques has revolutionized the segmentation process by automating it, thereby minimizing the requirement for manual intervention. This automation facilitates early detection and treatment of pancreatic cancer, offering significant benefits in terms of timely medical interventions.

VI. EVALUATION METRICS

The evaluation of segmentation techniques typically involves comparing the results obtained from the model with the ground truth segmentation obtained through manual annotations. The performance is evaluated using different evaluation metrics. The most commonly used metrics are the Dice similarity coefficient (DSC) and the Jaccard index (JI). DSC and JI [44], [70], [72], [73] are the similarity metrics that measures the similarity between ground truth and predicted segmentation results. DSC is defined in equation 1 as:

$$DSC = \frac{2 * |A \cap B|}{|A| + |B|}$$
(1)

Jaccard Index is defined in equation 2 as:

$$JI = \frac{|A \cap B|}{|A \cup B|} \tag{2}$$

where A and B represent the ground truth mask and the predicted mask, respectively. The numerator represents the number of pixels that are correctly segmented in both the ground truth and predicted masks, while the denominator denotes the cumulative pixel count in both masks. Both DSC and JI range from 0 to 1, with a value of 1 indicating perfect segmentation accuracy or resemblance in the ground truth and predicted masks. A low value of either metric indicates that the predicted mask is significantly different from the ground truth mask.

Accuracy [67] and [70] is also widely utilized for deep learning-based segmentation techniques. It measures the overall correctness of the segmentation results by calculating the proportion of correct pixels classified in the predicted segmentation. Accuracy is represented in equation 3

$$Accuracy = \frac{TP + TN}{TP + TN + FP + FN}$$
(3)

where TP (true positives) represents the count of pixels correctly classified as positive, TN (true negatives) represents the count of pixels correctly classified as negative, FP (false positives) represents the count of pixels incorrectly classified as positive, and FN (false negatives) represents the count of pixels incorrectly classified as negative.

Sensitivity (SI) or true positive rate (TPR) or recall [39], [60], is an evaluation metric used to measure the ability of a deep learning-based segmentation model to correctly detect positive samples. In the context of segmentation, positive samples refer to pixels or regions that belong to the target structure being segmented. Sensitivity is defined in equation 4 as:

$$Sensitivity = \frac{TP}{TP + FN} \tag{4}$$

Sensitivity ranges from 0 and 1, where 1 indicates the detection of all positive samples. A low sensitivity value indicates that the model is missing a significant number of positive samples, which can result in incomplete or inaccurate segmentation.

Specificity (SP) [45], [60] measures the ability of a segmentation model to correctly identify negative samples, which are samples that do not belong to the target structure being segmented. Equation 5 represents specificity as:

$$Specificity = \frac{TN}{TN + FP}$$
(5)

Specificity lies between 0 and 1, the 1 represents the identification of all negative samples. A low specificity value indicates that the model is misclassifying a significant number of negative samples as positive, which can result in incomplete or inaccurate segmentation.

Precision measures the accuracy of a segmentation model in correctly identifying positive samples, which are samples that belong to the target structure being segmented. Precision is defined in equation 6 as:

$$Precision = \frac{TP}{TP + FP} \tag{6}$$

Higher the value of accuracy, precision, and recall, the better the segmentation results [40], [73]. Hausdorff distance measures the accuracy of segmentation by calculating the maximum distance between the boundary points of the ground truth and predicted segmentation masks. A smaller Hausdorff distance indicates better segmentation accuracy, as it means that the boundary points of the predicted segmentation are near to those of the ground truth [68]. In pancreatic edge segmentation, for example, a smaller Hausdorff distance indicates better accuracy of the segmented pancreatic margin, as the predicted boundary points are closer to those of the ground truth. The Hausdorff distance is a valuable evaluation metric to complement other commonly used metrics such as DSC, Jaccard Index.

Area under the receiver operating characteristics curve (AUC) is used to evaluate the segmentation model by measuring its ability to differentiate between healthy and tumor tissues in medical images. The AUC is calculated as the area under the ROC curve, which plots the TPR against the FPR at different threshold values. A higher AUC value indicates better performance of the segmentation model [5].

The example of segmentation of the pancreas and PDAC mass is shown in FIGURE 14 with various DSC findings.

VII. EXPERIMENTAL DATASET

In pancreas and tumor segmentation, there is a limited availability of publicly accessible datasets. National Institutes of Health - The Cancer Imaging Archive (NIH-TCIA) dataset



FIGURE 14. Segmentation of PDAC and pancreas (a) Original CT image of different patients (b)Ground Truth (red denotes PDAC and green denotes pancreas) (c) Segmentation output (d) Visualization of PDAC in 3D with DSC value for various patients [63].

[74] is widely recognized as the most commonly used dataset for segmentation of the pancreas as it contains a sufficient amount of labeled data. This dataset comprises 82 CT images with contrast enhancement, each with a resolution of $512 \times$ 512 pixels. The slice thicknesses vary between 1.5 to 2.5 mm. The CT scans were acquired using multi-detector computed tomography (MDCT) scanners from Philips and Siemens, with a tube voltage of 120 kVp.

Another dataset from John Hopkins Medical Institutions (JHMI) [72] comprises 131 abdominal CT volumes with contrast enhancement, and each volume contains annotated pancreatic labels generated by human experts. The CT volumes in the JHMI dataset have a resolution of 512×512 pixels in the transverse plane, while the axial plane thickness, denoted as D, varies between 358 and 1121, providing a wider range of slice thickness compared to the NIH dataset.

Another widely used dataset for pancreatic tumor segmentation is the Medical Segmentation Decathlon (MSD) challenge dataset, that comprises of 420 CT scans of portal-venous phase patients with pancreatic tumors acquired from Memorial Sloan Kettering Cancer Center, New York, US.

There are also several private datasets used within institutions, but publicly available datasets are crucial for the researchers to evaluate the benefits and drawbacks of various segmentation methods. Comparing the performance of various methods using a similar dataset is more meaningful for the research community.

VIII. DISCUSSION

In this contemporary review, various pancreas and pancreatic tumor segmentation techniques are explored. Deep learning methods have emerged as a popular and promising approach, demonstrating superior performance in accurately segmenting the pancreas and pancreatic tumors than conventional segmentation methods. These methods leverage the power of neural networks to learn intricate patterns and features from large datasets, enabling them to capture complex anatomical structures and subtle tumor boundaries. The evaluation of segmentation performance varies across studies, with different metrics and benchmarks used for assessment.

Several segmentation techniques have been proposed for segmenting the pancreas into distinct groups, catering to different applications and imaging modalities.

The approaches for assessing segmentation performance varies in different ways. DSC parameter and Jaccard Index are largely adopted metrics to measure segmentation accuracy. In the realm of pancreas segmentation, atlas based method by Karasawa et al. [16] demonstrated superior results in terms of Dice Similarity Coefficient and Jaccard Index when compared to conventional multi-organ atlasbased segmentation methods. Their study utilized a dataset comprising 150 CT scans obtained from Nagoya University Hospital. Table 1 represents conventional multi-organ atlasbased segmentation methods.

However, recent advancements are particularly utilizing supervised machine learning and deep learning approaches such as U-Net and hybrid architectures, have achieved remarkable results for both pancreas and tumor segmentation. Many of these methods have achieved DSC values exceeding 80%. Table 2 shows the various pancreas segmentation methods and their performance using supervised machine learning and deep learning.

Noteworthy achievements include the U-Net with Trans-Deformer network proposed by Dai et al. [49], which achieved remarkable DSC results of 89.89% for the NIH dataset and 91% for the MSD challenge dataset for pancreas segmentation. The other method multiphase morphology-guided deep learning framework by Qureshi et al. [41], also which achieved a remarkable DSC of approximately 88.53%. Qiu et al. [37] demonstrated significant improvements in the Jaccard Index (JI) with a value of $76.26 \pm 5.01\%$ for pancreas segmentation, achieving better segmentation results for precision and recall as well.

The MBU-Net architecture [36] showcased exceptional precision $89.29 \pm 0.98\%$ and specificity $99.95 \pm 0.01\%$. Man et al. [32] employed deep Q learning and deformable U-Net, resulting in a higher recall of $86.93 \pm 4.88\%$. The V-Net architecture [42], achieving notable sensitivity 87.70% and precision 97.07%. Table 3 focuses on various deep learning based segmentation methods for pancreatic tumor and cyst. Few methods have performed both pancreas and pancreatic tumor/cyst segmentation [57], [58], [63], [65], [66], [72].

It is imperative for future research to standardize evaluation protocols and compare various segmentation methods using diverse datasets, ultimately advancing the field of pancreas and pancreatic tumor segmentation. However, it is worth noting the challenges in comparing pancreatic tumor

TABLE 1. Conventional pancreas segmentation methods.

Author	Modality	Method	Dataset	Results
Chen	CT	Image	40 MRI	DSC=67.2±15.5%
et al. ,		registration	scans	
2016		with label		
		fusion		
		method		
Karasawa	I CT	Atlas	150 CT	DSC= 76.3 ±
et al. ,		selection	scans	16.4%, JI= 63.9 \pm
2017		based	from	17.1%
		on vessel	Nagoya	
		structure	Uni-	
			versity	
			Hospi-	
			tal	
Saito	СТ	Joint	140 CT	DSC= 74.4 \pm
et al.,		optimization	scans	20.2%, JI= 62.3 \pm
2016		algorithm		19.5
Oda et	CT	EM algorithm	147 CT	Dice overlap=
al. [19],		with	scans	$75.1 \pm 15.4\%$, JI=
2016		graph-cut		62.1±16.6%
		optimization		
Tong	CT	DDLS Algo-	150 CT	DSC= $65.5 \pm$
et al.		rithm	scans	17.8%
, [14],			from	
2015			Nagoya	
			Uni-	
			versity	
			Hospi-	
			tal	
Wolz et	СТ	Hierarchical	150 CT	$DSC=69.6 \pm$
al. [15],		atlas	scans	16.7%, JI= 55.5 \pm
2013		registration	from	17.1%, Recall =
		integrated	Nagoya	$67.9 \pm 18.2\%$, Pr =
		with	Uni-	$74.1 \pm 17.1\%$
		weighting	versity	
		scheme	Hospi-	
	07	G	tal	
Chu et	CT	Spatially	100 CT	$DSC = 69.1 \pm 15.0\%$
al. [17],		divided	scans	15.3%, JI = $54.6%$
2013		probabilisite		
		atlases		

segmentation methods, as achieving high DSC values on private datasets is feasible, but researchers face difficulties in achieving comparable results on the MSD challenge dataset.

In a recent study by Du et al. [56], they achieved a noteworthy DSC of 68.03% for pancreatic tumor segmentation on a private dataset focusing on small size tumor. Mahmoudi et al. [63] achieved a DSC of about 60. 6% on the MSD challenge dataset. Furthermore, Alves et al. [61] focused on tumor sizes smaller than 2cm obtaining better results for Area Under the Curve (AUC) about 0. 876. The scarcity of labeled pancreatic tumor datasets has impeded the development of segmentation techniques, leading many researchers to rely on privately available datasets and the MSD challenge dataset for their studies.

In general, there has been a notable improvement in the accuracy assessment of pancreas segmentation methods. However, there remains significant variation in evaluation criteria, and datasets used across different research articles, leading to a lack of consistency. Deep learning methods have gained substantial popularity in pancreas segmentation research, while research specifically focused on pancreatic

TABLE 2. Pancreas segmentation methods using supervised machine learning and deep learning.

Author	Modality	Method	Dataset	Results
Dai [49], 2023	СТ	Trans-Deformer Network	NIH TCIA pancreas dataset & MSD challenge dataset	NIH TCIA pancreas dataset DSC=89.89 \pm 1.82%, MSD challenge dataset DSC= 91.22 \pm 1.37%
Qiu [37], 2023	СТ	Cascaded multiscale feature calibration U-Net (CMFCUNet)	NIH TCIA pancreas dataset	DSC = 86.30±4.03%, JI = 76.26±5.01%, Pr= 85.91±3.97%, Recall= 86.85±5.23%
He et al. [38], 2023	СТ	U-Netmer	MSD Challenge dataset	DSC = 79.42 \pm 7.59%, JI = 66.46 \pm 9.63%, Accuracy=99.47 \pm 0.23%, SI= 78.49 \pm 11.77%, SP=99.78 \pm 0.11%
Quereshi [41], 2022	СТ	VGG net& U-Net	NIH TCIA pancreas dataset	DSC = 88.53%
Haipeng Chen [35], 2022	СТ	TVMS-Net	NIH TCIA pancreas dataset & MSD challenge dataset	NIH TCIA pancreas dataset- DSC=85.19 \pm 4.73%, Pr=86.09 \pm 5.93%, Recall=84.58 \pm 8.09% IOU=74.19 \pm 7.27% & MSD Challenge dataset-DSC=76.6 \pm 7.3%, Pr=87.7 \pm 8.3%, Recall=69.2 \pm 12.8%, IOU=62.6 \pm 9.3%
Mei-Ling Huang [36], 2022	СТ	MBU-Net (Mobile U- Net)	NIH TCIA pancreas dataset	DSC=82.87 \pm 1.00%, JI=70.97 \pm 1.39%, Pr=89.29 \pm 0.98%, SP=99.95 \pm 0.01%, Recall=77.37 \pm 1.41%
Moritz Knolle [75], 2021	СТ	MoNet	MSD challenge Dataset and independent validation set (IVD)	Dice Score for MSD dataset= $78\pm9.0\%$ Hausdorff Distance for MSD dataset = 1.78 ± 0.61 mm, Dice Score for IVD Dataset $70\pm10\%$
Yuan Wang [39], 2021	СТ	V-mesh network	NIH TCIA pancreas dataset	DSC=87.4 \pm 6.8%, SI=87.7 \pm 7.9%
Jun Li [40], 2021	СТ	PBR-UNet	NIH TCIA pancreas dataset	DSC=84.19 \pm 5.73%, HD=3.19 \pm 0.40 mm, RMSE=3.60 \pm 2.57 mm, Recall=82.23 \pm 9.14%, Pr=81.44 \pm 7.53%, IOU= 72.62 \pm 9.09%
T. G. W. Boers [76], 2020	СТ	3D interactive U-NET	Private Dataset	DSC=78.1±8.7%
Bharat Giddwani [42], 2020	СТ	V-NET	NIH TCIA pancreas dataset	DSC=83.31%, SI=87.70%, Pr=97.07%
Juan Mo [43], 2020	СТ	CNN	NIH TCIA pancreas dataset	DSC=82.47±5.50%
Mizuho Nishio [44], 2020	СТ	Deep U-Net	NIH TCIA pancreas dataset	DSC=78.9 \pm 8.3%, JI=65.8 \pm 10.3%, SI=76.2 \pm 12%, SP=1.000 \pm 0%
Zhuotun Zhu [45], 2019	СТ	Neural architecture search	NIH TCIA pancreas dataset & MSD challenge dataset	For NIH dataset: pancreas DSC= $85.15 \pm 4.55\%$, For MSD challenge: pancreas DSC= $79.94 \pm 8.85\%$
JIA Fan [46], 2019	СТ	Regularized U-Net (RUNet)	NIH TCIA pancreas dataset	Accuracy= 99.43%, mean intersection over union (mIoU)= 75.73, DSC= 68.45%
Ningning Zhao [47], 2019	СТ	U-Net	NIH TCIA pancreas dataset	DSC=84.10 \pm 4.91%, JI=72.86 \pm 6.89%, Pr=83.60 \pm 5.85%, Recall=85.33 \pm 8.24%
S. Liu [48], 2019	СТ	U-Net	NIH TCIA pancreas dataset	DSC=84.10 \pm 4.91%, JI=72.86 \pm 6.89%, Pr=83.60 \pm 5.85%, Recall 85.33 \pm 8.24%
Man [32], 2019	СТ	Deep Q Learning & de- formable U-Net	NIH TCIA pancreas dataset	DSC= $86.93 \pm 4.92\%$, Recall= $86.91 \pm 4.88\%$
Xue [33], 2019	СТ	3D FCN	NIH TCIA pancreas dataset & Fujian Medical University dataset	For NIH TCIA dataset DSC = $85.9 \pm 5.1\%$, JI = $75.7 \pm 7.6\%$, Pr = $87.6 \pm 4.7\%$, Recall = $85.2 \pm 8.9\%$
Oktay [31], 2018	СТ	Attention gate model with U-NET	NIH TCIA pancreas dataset	DSC=83.1 \pm 3.8%, Pr=82.5 \pm 7.3%, Recall=84.0 \pm 5.3%
Roth et al. [30], 2018	СТ	Spatial aggregation us-	NIH TCIA pancreas	DSC=81.27 \pm 6.27%
Zhu et al. [29], 2018	СТ	ResDSN Coarse to Fine model	NIH TCIA pancreas	DSC=83.18 ± 6.02%
Yu et al. [26], 2018	СТ	Reccurent saliency Transformation Network	NIH TCIA pancreas dataset	DSC=84.50 ± 4.97%
Zhou et al. [25], 2017	СТ	Coarse to fine segmen- tation	NIH TCIA dataset	DSC= 82.37 ± 5.68%
MinFu [73], 2016	СТ	Richer convolutional features network (RCF)	Private dataset from General Surgery Department of Peking Union Medical College Hospital	DSC=76.36 ± 17.96%, Pr=77.36 ± 17.96%, Recall= 79.12 ± 16.27%, JI=63.72 ± 17.05%

TABLE 3. Pancreatic tumor and cyst segmentation methods.

Author	Modality	Method	Dataset	Results
Du et al. [56],	СТ	Multi-scale	CT images of 419	DSC= 68.03% & Jaccard Index= 59.31%
2023		channel attention	patients from affiliated	
		U-Net	Hospital of Qingdao	
			University	
Qi Li et al. [57], 2023	СТ	3D FCN	MSD challenge dataset	For pancreas: DSC= 85.06±7.71% , For Tumor: DSC= 59.16±28.12%
Chen et al. [59],	СТ	Unified Tumor	Shengjing Hospital of	DSC = 70.2%
2023		Transformer	China Medical Univer-	
		(UniT) model	sity	
Seak Oh et al.	endoscopic ultra-	Attention U-Net	EUS Dataset A- 52 pa-	Dataset A- Accuracy= $98.3 \pm 1.2\%$,
[60], 2022	sound (EUS)		tients with 57 images	SP=99.1 \pm 0.9%, SI=79.7 \pm 32.7%,
			from the Gil Medical	DSC=79.4±32.6%, IOU=7.41(0.308) & Dataset
			Center & EUS Dataset	B- Accuracy= $97.2 \pm 3.7\%$, SP= $98.9 \pm 1.4\%$,
			B- 59 patients with 364	SI=72.3 \pm 32.2%, DSC=69.1 \pm 28.3%,
			images from the Sever-	IOU=58.7±27.6%
			ance Hospital	
Natalia Alves et	CT	nn-UNet	MSD Challenge	AUC = 0.914 of the entire external test set and
al. [61], 2022	61177		Dataset	AUC = 0.876 for the tumors less than 2 cm in size
Ke Si et al. [62],	CT	RESNET & U-	319 patients from	AUC=0.871, and Accuracy of PDAC=87.6,
2021		NEI	Znejiang University	Accuracy of IPMIN=100%, Accuracy of
			China with 142.045	SCIN=81.3%, Average accuracy=82.7% on
			dunamia CT imagas of	an independent testing dataset of 547 patient.
			the abdomen	Computation time =18.6 s per patient
Tahereh	СТ	Texture attention	MSD challenge dataset	For pancreas: Dice Global=72.7% Dice per
Mahmoudi	0.1	U-Net(TAU-Net)	nibb enalienge aaaset	Case= $73.8 \pm 8\%$. Recall= $70.9 \pm 21\%$. Pr=
et al. [63], 2021		0 1.00(1110 1.00)		$70.1\pm$ 1%, For PDAC mass: Dice Global
				=60.6%, Dice per case=57.3 \pm 15%, Recall=
				$78.0\pm9\%$, Pr= $57.8\pm23\%$
Yuhei Iwasa et al.	Contrast	U-Net	Video images obtained	JI=70%
[64], 2021	Enhanced-		from National Taiwan	
	Endoscopic		University Hospital,	
	Ultrasound		Bei-Hu Branch,	
			and Gifu University	
	0.00		Hospital	
Li et al. $[65],$	CT	Position guided	MSD Challenge dataset	DSC for pancreas = $77.01 \pm 10.47\%$, DSC for
2020		deformable		Tumors = $50.12 \pm 30.86\%$
		UNet (PGD-		
Alzbeta Tureck-	СТ	V-Net	MSD challenge dataset	Dice Score for pancreas- 81 22% and Dice
ova et al [66]		V-1101	wist chantenge dataset	Score for Tumor= 52 99%
2020				
Feng Jiang et al.	СТ	DLU-Net	MSD challenge dataset	MSD Dataset: DLU-Net Accuracy=97.25%, Pr=
[67], 2020			and Changhai Hospital	88.01%, SI= 94.53%, SP= 97.73%
			Dataset	
Ying Liang et al.	MRI	CNN	40 patients with PDAC	DSC=71± 8%, HD=7.3±2.72mm
[68], 2020				
H. Zheng et al.	MRI/CT	U-NET	1. MRI Dataset from	MRI Dataset:DSC=73.88%, Accuracy=99.86%,
[70], 2020			Changai Hospital (20	Pr=86.06%, Recall=65.42%, NIH CT
			patients) with PDAC.	Dataset: $DSC = 84.3/\%$, Accuracy=99.95%,
			2. NIH TCIA pancreas	PT=63.10%, Kecall= 80.26%
Zhuotun Zhu at	СТ	Segmentation	Total 430 CT scope of	DSC normal pancreas- 86.0±8.6% Abnormal
a] [58] 2010		for-classification	which 136 cases having	pancreas $-81.0\pm10.8\%$ Tumor $-57.3\pm29.1\%$
ai. [30], 2017		network	PDAC and 303 cases	$\begin{vmatrix} \text{panereds} = 01.0 \pm 10.0\%, \text{fullor} = 57.5 \pm 20.1\%, \\ \text{SI} = 92.7\% \text{SP} = 99\% \end{aligned}$
		network	are normal cases	51 - <i>72.1 /0</i> , 51 - <i>77 /0</i>
Guo et al. [71]	СТ	U-Net with	15 patients with pan-	DSC=83.2±7.8%
2018		LOGISMOS	creatic tumor having 51	
		approach	arterial phase CT scans	
Yuyin Zhou et al.	СТ	Deep FCN	131-CT scans	pancreas DSC=79.23±9.72%, cyst
[72], 2017				DSC=63.44±27.710%

tumor segmentation is relatively limited. However, until now, no method has obtained satisfactory performance, specifically for tumor segmentation.

Nevertheless, it is worth noting that the segmentation results have exhibited significant improvement over time.

This advancement is crucial for future progress in the field. Moving forward, the focus of research will shift towards the development of new or hybrid methods. The pancreas is well known for its significant diversity in form, size, and position while constituting just a tiny proportion of the total

TABLE 4. Research gaps and future directions.

Research Gaps	Future Research Directions
Pancreatic tumor with a size	Improved AI based methods
less than 2cm is difficult to	will help to precisely segment
segment.	small size tumors.
The segmentation	It is crucial to focus on AI
performance for pancreatic	based models that can im-
cancer is currently	prove the segmentation per-
inadequate.	formance.
Insufficient amount of anno- tated medical data.	To lessen the dependency on annotated data, alternative learning strategies, including semi-supervised learning and transfer learning, should be
Complexity of AI algorithms	To improve interpretability and transparency, future research should concentrate on creating AI models that are easy to understand.

CT volume of 0. 5%. As a result, segmenting the pancreatic tumor remains tough. Table 4 represents research gaps and future directions in pancreatic cancer segmentation.

IX. SUMMARY

Automatic segmentation of pancreas and pancreatic tumor is the crucial topic of this in-depth review, which highlights the substantial improvements and difficulties in medical imaging. The review aims to offer a precise overview of segmentation methods and advancements made in the past ten years. A systematic search across various databases and sources is conducted using inclusion and exclusion criteria to perform this detailed review. The segmentation methods for the pancreas are divided into different approaches, including conventional methods that emphasize traditional registration and atlas-based segmentation techniques. Supervised machine learning and deep learning methods focus on CNN, U-Net, and modifications to U-Net architectures. Lastly, unsupervised machine learning concentrates on clustering techniques. Pancreatic tumor segmentation similarly centers around deep learning-based architectures, including variations in the U-Net model. Advancement in deep learning-based methods can overcome the challenges and help to segment the pancreas and tumor segmentation efficiently. The segmentation methods have shown substantial improvement over the years. Though segmentation of pancreatic tumors is quite difficult, the review also highlights the current research gaps that can help researchers focus on developing more effective and accurate segmentation.

X. CONCLUSION AND FUTURE RESEARCH DIRECTIONS

In conclusion, the review paper provides a detailed overview and the advancements in pancreas and pancreatic tumor segmentation. Significant progress can be observed in the segmentation of the pancreas and tumor using deep learning methods. Promising results are achieved in the segmentation of the pancreas using deep learning-based methods. The limiting factor is the publicly available labeled datasets which has led researchers to use private datasets for the implementation of pancreatic tumor segmentation. Pancreas segmentation is more extensively explored than pancreatic tumor segmentation due to the different forms and size of tumor. Future research should focus on developing new approaches and enhancing the performance of pancreas and pancreatic tumor segmentation. Standardizing evaluation protocols and utilizing diverse datasets will be crucial in further advancing the field of pancreas and pancreatic tumor segmentation. Additionally, efforts should be made to improve the accessibility of labeled datasets to facilitate research and promote reproducibility. Overall, the advancements made in pancreas and pancreatic tumor segmentation hold great promise for improving diagnosis, treatment planning, and monitoring of pancreatic diseases. Continued research and collaboration in this field will lead to further breakthroughs, ultimately benefiting patients and advancing the medical imaging analysis field.

REFERENCES

- [1] H. Sung, J. Ferlay, R. L. Siegel, M. Laversanne, I. Soerjomataram, A. Jemal, and F. Bray, "Global cancer statistics 2020: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries," *CA, A Cancer J. Clinicians*, vol. 71, no. 3, pp. 209–249, May 2021.
- [2] C. Luchini, P. Capelli, and A. Scarpa, "Pancreatic ductal adenocarcinoma and its variants," *Surgical Pathol. Clinics*, vol. 9, no. 4, pp. 547–560, Dec. 2016.
- [3] J. D. Mizrahi, R. Surana, J. W. Valle, and R. T. Shroff, "Pancreatic cancer," *Lancet*, vol. 395, no. 10242, pp. 2008–2020, 2020.
- [4] A. L. Blackford, M. I. Canto, A. P. Klein, R. H. Hruban, and M. Goggins, "Recent trends in the incidence and survival of stage 1A pancreatic cancer: A surveillance, epidemiology, and end results analysis," *JNCI, J. Nat. Cancer Inst.*, vol. 112, no. 11, pp. 1162–1169, Nov. 2020.
- [5] K.-L. Liu, T. Wu, P.-T. Chen, Y. M. Tsai, H. Roth, M.-S. Wu, W.-C. Liao, and W. Wang, "Deep learning to distinguish pancreatic cancer tissue from non-cancerous pancreatic tissue: A retrospective study with cross-racial external validation," *Lancet Digit. Health*, vol. 2, no. 6, pp. e303–e313, Jun. 2020.
- [6] X. Zhang, Y. Zhang, G. Zhang, X. Qiu, W. Tan, X. Yin, and L. Liao, "Deep learning with radiomics for disease diagnosis and treatment: Challenges and potential," *Frontiers Oncol.*, vol. 12, Feb. 2022, Art. no. 773840.
- [7] Z. Yang, "Pancreas segmentation in abdominal CT scans using inter-/intraslice contextual information with a cascade neural network," in *Proc. 41st Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC)*, Jul. 2019, pp. 5937–5940.
- [8] M. A. Atkinson, M. Campbell-Thompson, I. Kusmartseva, and K. H. Kaestner, "Organisation of the human pancreas in health and in diabetes," *Diabetologia*, vol. 63, no. 10, pp. 1966–1973, Oct. 2020.
- [9] K. J. Mortelé, T. C. Rocha, J. L. Streeter, and A. J. Taylor, "Multimodality imaging of pancreatic and biliary congenital anomalies," *RadioGraphics*, vol. 26, no. 3, pp. 715–731, May 2006.
- [10] V. Gulshan, L. Peng, M. Coram, M. C. Stumpe, D. Wu, A. Narayanaswamy, S. Venugopalan, K. Widner, T. Madams, J. Cuadros, R. Kim, R. Raman, P. C. Nelson, J. L. Mega, and D. R. Webster, "Development and validation of a deep learning algorithm for detection of diabetic retinopathy in retinal fundus photographs," *JAMA*, vol. 316, no. 22, p. 2402, Dec. 2016.
- [11] K. Yasaka, H. Akai, O. Abe, and S. Kiryu, "Deep learning with CNN showed high diagnostic performance in differentiation of liver masses at dynamic CT," *Radiology*, vol. 286, pp. 887–896, Mar. 2018.
- [12] A. Esteva, B. Kuprel, R. A. Novoa, J. Ko, S. M. Swetter, H. M. Blau, and S. Thrun, "Dermatologist-level classification of skin cancer with deep neural networks," *Nature*, vol. 542, no. 7639, pp. 115–118, Feb. 2017.
- [13] J. Shen, T. Baum, C. Cordes, B. Ott, T. Skurk, H. Kooijman, E. J. Rummeny, H. Hauner, B. H. Menze, and D. C. Karampinos, "Automatic segmentation of abdominal organs and adipose tissue compartments in water-fat MRI: Application to weight-loss in obesity," *Eur. J. Radiol.*, vol. 85, no. 9, pp. 1613–1621, Sep. 2016.

- [14] T. Tong, R. Wolz, Z. Wang, Q. Gao, K. Misawa, M. Fujiwara, K. Mori, J. V. Hajnal, and D. Rueckert, "Discriminative dictionary learning for abdominal multi-organ segmentation," *Med. Image Anal.*, vol. 23, no. 1, pp. 92–104, Jul. 2015.
- [15] R. Wolz, C. Chu, K. Misawa, M. Fujiwara, K. Mori, and D. Rueckert, "Automated abdominal multi-organ segmentation with subject-specific atlas generation," *IEEE Trans. Med. Imag.*, vol. 32, no. 9, pp. 1723–1730, Sep. 2013.
- [16] K. Karasawa, M. Oda, T. Kitasaka, K. Misawa, M. Fujiwara, C. Chu, G. Zheng, D. Rueckert, and K. Mori, "Multi-atlas pancreas segmentation: Atlas selection based on vessel structure," *Med. Image Anal.*, vol. 39, pp. 18–28, Jul. 2017.
- [17] C. Chu, M. Oda, T. Kitasaka, K. Misawa, M. Fujiwara, Y. Hayashi, Y. Nimura, D. Rueckert, and K. Mori, "Multi-organ segmentation based on spatially-divided probabilistic atlas from 3D abdominal CT images," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2013, pp. 165–172.
- [18] A. Saito, S. Nawano, and A. Shimizu, "Joint optimization of segmentation and shape prior from level-set-based statistical shape model, and its application to the automated segmentation of abdominal organs," *Med. Image Anal.*, vol. 28, pp. 46–65, Feb. 2016.
- [19] M. Oda, N. Shimizu, K. Karasawa, Y. Nimura, T. Kitasaka, K. Misawa, M. Fujiwara, D. Rueckert, and K. Mori, "Regression forest-based atlas localization and direction specific atlas generation for pancreas segmentation," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Athens, Greece: Springer, Oct. 2016, pp. 556–563.
- [20] H. R. Roth, A. Farag, L. Lu, E. B. Turkbey, and R. M. Summers, "Deep convolutional networks for pancreas segmentation in CT imaging," *Proc. SPIE*, vol. 9413, Mar. 2015, Art. no. 94131G.
- [21] H. Chen, Q. Dou, L. Yu, J. Qin, and P.-A. Heng, "VoxResNet: Deep voxelwise residual networks for brain segmentation from 3D MR images," *NeuroImage*, vol. 170, pp. 446–455, Apr. 2018.
- [22] V. Badrinarayanan, A. Kendall, and R. Cipolla, "SegNet: A deep convolutional encoder-decoder architecture for image segmentation," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 39, no. 12, pp. 2481–2495, Dec. 2017.
- [23] X. Li, Q. Dou, H. Chen, C.-W. Fu, X. Qi, D. L. Belavý, G. Armbrecht, D. Felsenberg, G. Zheng, and P.-A. Heng, "3D multi-scale FCN with random modality voxel dropout learning for intervertebral disc localization and segmentation from multi-modality MR images," *Med. Image Anal.*, vol. 45, pp. 41–54, Apr. 2018.
- [24] O. Ronneberger, P. Fischer, and T. Brox, "U-Net: Convolutional networks for biomedical image segmentation," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2015, pp. 234–241.
- [25] Y. Zhou, L. Xie, W. Shen, Y. Wang, E. K. Fishman, and A. L. Yuille, "A fixed-point model for pancreas segmentation in abdominal CT scans," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2017, pp. 693–701.
- [26] Q. Yu, L. Xie, Y. Wang, Y. Zhou, E. K. Fishman, and A. L. Yuille, "Recurrent saliency transformation network: Incorporating multi-stage visual cues for small organ segmentation," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, Jun. 2018, pp. 8280–8289.
- [27] S. Xie and Z. Tu, "Holistically-nested edge detection," in *Proc. IEEE Int. Conf. Comput. Vis. (ICCV)*, Dec. 2015, pp. 1395–1403.
- [28] H. R. Roth, L. Lu, A. Farag, A. Sohn, and R. M. Summers, "Spatial aggregation of holistically-nested networks for automated pancreas segmentation," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2016, pp. 451–459.
- [29] Z. Zhu, Y. Xia, W. Shen, E. Fishman, and A. Yuille, "A 3D coarse-tofine framework for volumetric medical image segmentation," in *Proc. Int. Conf. 3D Vis. (3DV)*, Sep. 2018, pp. 682–690.
- [30] H. R. Roth, L. Lu, N. Lay, A. P. Harrison, A. Farag, A. Sohn, and R. M. Summers, "Spatial aggregation of holistically-nested convolutional neural networks for automated pancreas localization and segmentation," *Med. Image Anal.*, vol. 45, pp. 94–107, Apr. 2018.
- [31] O. Oktay, J. Schlemper, L. Le Folgoc, M. Lee, M. Heinrich, K. Misawa, K. Mori, S. McDonagh, N. Y Hammerla, B. Kainz, B. Glocker, and D. Rueckert, "Attention U-Net: Learning where to look for the pancreas," 2018, arXiv:1804.03999.
- [32] Y. Man, Y. Huang, J. Feng, X. Li, and F. Wu, "Deep Q learning driven CT pancreas segmentation with geometry-aware U-Net," *IEEE Trans. Med. Imag.*, vol. 38, no. 8, pp. 1971–1980, Aug. 2019.

- [33] J. Xue, K. He, D. Nie, E. Adeli, Z. Shi, S.-W. Lee, Y. Zheng, X. Liu, D. Li, and D. Shen, "Cascaded MultiTask 3-D fully convolutional networks for pancreas segmentation," *IEEE Trans. Cybern.*, vol. 51, no. 4, pp. 2153–2165, Apr. 2021.
- [34] A. Farag, L. Lu, H. R. Roth, J. Liu, E. Turkbey, and R. M. Summers, "A bottom-up approach for pancreas segmentation using cascaded superpixels and (deep) image patch labeling," *IEEE Trans. Image Process.*, vol. 26, no. 1, pp. 386–399, Jan. 2017.
- [35] H. Chen, Y. Liu, Z. Shi, and Y. Lyu, "Pancreas segmentation by two-view feature learning and multi-scale supervision," *Biomed. Signal Process. Control*, vol. 74, Apr. 2022, Art. no. 103519.
- [36] M.-L. Huang and Y.-Z. Wu, "Semantic segmentation of pancreatic medical images by using convolutional neural network," *Biomed. Signal Process. Control*, vol. 73, Mar. 2022, Art. no. 103458.
- [37] C. Qiu, Y. Song, Z. Liu, J. Yin, K. Han, and Y. Liu, "CMFCUNet: Cascaded multi-scale feature calibration UNet for pancreas segmentation," *Multimedia Syst.*, vol. 29, no. 2, pp. 871–886, Apr. 2023.
- [38] S. He, R. Bao, P. Ellen Grant, and Y. Ou, "U-netmer: U-Net meets transformer for medical image segmentation," 2023, arXiv:2304.01401.
- [39] Y. Wang, G. Gong, D. Kong, Q. Li, J. Dai, H. Zhang, J. Qu, X. Liu, and J. Xue, "Pancreas segmentation using a dual-input v-mesh network," *Med. Image Anal.*, vol. 69, Apr. 2021, Art. no. 101958.
- [40] J. Li, X. Lin, H. Che, H. Li, and X. Qian, "Pancreas segmentation with probabilistic map guided bi-directional recurrent UNet," *Phys. Med. Biol.*, vol. 66, no. 11, Jun. 2021, Art. no. 115010.
- [41] T. A. Qureshi, C. Lynch, L. Azab, Y. Xie, S. Gaddam, S. J. Pandol, and D. Li, "Morphology-guided deep learning framework for segmentation of pancreas in computed tomography images," *J. Med. Imag.*, vol. 9, no. 2, Apr. 2022, Art. no. 024002.
- [42] B. Giddwani, H. Tekchandani, and S. Verma, "Deep dilated V-Net for 3D volume segmentation of pancreas in CT images," in *Proc. 7th Int. Conf. Signal Process. Integr. Netw. (SPIN)*, Feb. 2020, pp. 591–596.
- [43] J. Mo, L. Zhang, Y. Wang, and H. Huang, "Iterative 3D feature enhancement network for pancreas segmentation from CT images," *Neural Comput. Appl.*, vol. 32, no. 16, pp. 12535–12546, Aug. 2020.
- [44] M. Nishio, S. Noguchi, and K. Fujimoto, "Automatic pancreas segmentation using coarse-scaled 2D model of deep learning: Usefulness of data augmentation and deep U-Net," *Appl. Sci.*, vol. 10, no. 10, p. 3360, May 2020.
- [45] Z. Zhu, C. Liu, D. Yang, A. Yuille, and D. Xu, "V-NAS: Neural architecture search for volumetric medical image segmentation," in *Proc. Int. Conf. 3D Vis. (3DV)*, Sep. 2019, pp. 240–248.
- [46] J. Fan and X.-C. Tai, "Regularized UNet for automated pancreas segmentation," in *Proc. 3rd Int. Symp. Image Comput. Digit. Med.*, Aug. 2019, pp. 113–117.
- [47] N. Zhao, N. Tong, D. Ruan, and K. Sheng, "Fully automated pancreas segmentation with two-stage 3D convolutional neural networks," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2019, pp. 201–209.
- [48] S. Liu, X. Yuan, R. Hu, S. Liang, S. Feng, Y. Ai, and Y. Zhang, "Automatic pancreas segmentation via coarse location and ensemble learning," *IEEE Access*, vol. 8, pp. 2906–2914, 2020.
- [49] S. Dai, Y. Zhu, X. Jiang, F. Yu, J. Lin, and D. Yang, "TD-Net: Trans-deformer network for automatic pancreas segmentation," *Neurocomputing*, vol. 517, pp. 279–293, Jan. 2023.
- [50] K. P. Sinaga and M.-S. Yang, "Unsupervised K-means clustering algorithm," *IEEE Access*, vol. 8, pp. 80716–80727, 2020.
- [51] I. T. Jolliffe and J. Cadima, "Principal component analysis: A review and recent developments," *Phil. Trans. Roy. Soc. A, Math., Phys. Eng. Sci.*, vol. 374, no. 2065, Apr. 2016, Art. no. 20150202.
- [52] D. Lee and H. S. Seung, "Algorithms for non-negative matrix factorization," in Proc. Adv. Neural Inf. Process. Syst., vol. 13, 2000, pp. 556–562.
- [53] R. R. Roy and G. S. A. Mala, "An improved K-means clustering for segmentation of pancreatic tumor from CT images," *IETE J. Res.*, vol. 69, no. 7, pp. 3966–3973, Sep. 2023.
- [54] S. Jain, G. Sikka, and R. Dhir, "An automatic cascaded approach for pancreas segmentation via an unsupervised localization using 3D CT volumes," *Multimedia Syst.*, vol. 29, no. 4, pp. 2337–2349, Aug. 2023.
- [55] Y. Tang, J. Huang, W. Pedrycz, B. Li, and F. Ren, "A fuzzy clustering validity index induced by triple center relation," *IEEE Trans. Cybern.*, vol. 53, no. 8, pp. 5024–5036, Aug. 2023.

- [56] Y. Du, X. Zuo, S. Liu, D. Cheng, J. Li, M. Sun, X. Zhao, H. Ding, and Y. Hu, "Segmentation of pancreatic tumors based on multi-scale convolution and channel attention mechanism in the encoder-decoder scheme," *Med. Phys.*, to be published.
- [57] Q. Li, X. Liu, Y. He, D. Li, and J. Xue, "Temperature guided network for 3D joint segmentation of the pancreas and tumors," *Neural Netw.*, vol. 157, pp. 387–403, Jan. 2023.
- [58] Z. Zhu, Y. Xia, L. Xie, E. K. Fishman, and A. L. Yuille, "Multi-scale coarse-to-fine segmentation for screening pancreatic ductal adenocarcinoma," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2019, pp. 3–12.
- [59] J. Chen, Y. Xia, J. Yao, K. Yan, J. Zhang, L. Lu, F. Wang, B. Zhou, and M. Qiu, "Towards a single unified model for effective detection, segmentation, and diagnosis of eight major cancers using a large collection of CT scans," 2023, arXiv:2301.12291.
- [60] S. Oh, Y.-J. Kim, Y.-T. Park, and K.-G. Kim, "Automatic pancreatic cyst lesion segmentation on EUS images using a deep-learning approach," *Sensors*, vol. 22, no. 1, p. 245, Dec. 2021.
- [61] N. Alves, M. Schuurmans, G. Litjens, J. S. Bosma, J. Hermans, and H. Huisman, "Fully automatic deep learning framework for pancreatic ductal adenocarcinoma detection on computed tomography," *Cancers*, vol. 14, no. 2, p. 376, Jan. 2022.
- [62] K. Si, Y. Xue, X. Yu, X. Zhu, Q. Li, W. Gong, T. Liang, and S. Duan, "Fully end-to-end deep-learning-based diagnosis of pancreatic tumors," *Theranostics*, vol. 11, no. 4, pp. 1982–1990, 2021.
- [63] T. Mahmoudi, Z. M. Kouzahkanan, A. R. Radmard, R. Kafieh, A. Salehnia, A. H. Davarpanah, H. Arabalibeik, and A. Ahmadian, "Segmentation of pancreatic ductal adenocarcinoma (PDAC) and surrounding vessels in CT images using deep convolutional neural networks and texture descriptors," *Sci. Rep.*, vol. 12, no. 3092, pp. 1–14, 2022.
- [64] Y. Iwasa, T. Iwashita, Y. Takeuchi, H. Ichikawa, N. Mita, S. Uemura, M. Shimizu, Y.-T. Kuo, H.-P. Wang, and T. Hara, "Automatic segmentation of pancreatic tumors using deep learning on a video image of contrastenhanced endoscopic ultrasound," *J. Clin. Med.*, vol. 10, no. 16, p. 3589, Aug. 2021.
- [65] Z. Li, H. Pan, Y. Zhu, and A. K. Qin, "PGD-UNet: A position-guided deformable network for simultaneous segmentation of organs and tumors," in *Proc. Int. Joint Conf. Neural Netw. (IJCNN)*, Jul. 2020, pp. 1–8.
- [66] A. Turečková, T. Tureček, Z. Komínková Oplatková, and A. Rodríguez-Sánchez, "Improving CT image tumor segmentation through deep supervision and attentional gates," *Frontiers Robot. AI*, vol. 7, p. 106, Aug. 2020.
- [67] F. Jiang, X. Zhi, X. Ding, W. Tong, and Y. Bian, "DLU-Net for pancreatic cancer segmentation," in *Proc. IEEE Int. Conf. Bioinf. Biomed. (BIBM)*, Dec. 2020, pp. 1024–1028.
- [68] Y. Liang, D. Schott, Y. Zhang, Z. Wang, H. Nasief, E. Paulson, W. Hall, P. Knechtges, B. Erickson, and X. A. Li, "Auto-segmentation of pancreatic tumor in multi-parametric MRI using deep convolutional neural networks," *Radiotherapy Oncol.*, vol. 145, pp. 193–200, Apr. 2020.
- [69] L. C. Chu, "Application of deep learning to pancreatic cancer detection: Lessons learned from our initial experience," J. Amer. College Radiol., vol. 16, no. 9, pp. 1338–1342, Sep. 2019.
- [70] H. Zheng, Y. Chen, X. Yue, C. Ma, X. Liu, P. Yang, and J. Lu, "Deep pancreas segmentation with uncertain regions of shadowed sets," *Magn. Reson. Imag.*, vol. 68, pp. 45–52, May 2020.
- [71] Z. Guo, "Deep LOGISMOS: Deep learning graph-based 3D segmentation of pancreatic tumors on CT scans," in *Proc. IEEE 15th Int. Symp. Biomed. Imag. (ISBI)*, Apr. 2018, pp. 1230–1233.
- [72] Y. Zhou, L. Xie, E. K. Fishman, and A. L. Yuille, "Deep supervision for pancreatic cyst segmentation in abdominal CT scans," in *Proc. Int. Conf. Med. Image Comput. Computer-Assisted Intervent.* Cham, Switzerland: Springer, 2017, pp. 222–230.
- [73] M. Fu, W. Wu, X. Hong, Q. Liu, J. Jiang, Y. Ou, Y. Zhao, and X. Gong, "Hierarchical combinatorial deep learning architecture for pancreas segmentation of medical computed tomography cancer images," *BMC Syst. Biol.*, vol. 12, no. S4, pp. 119–127, Apr. 2018.
- [74] H. R. Roth, L. Lu, A. Farag, H.-C. Shin, J. Liu, E. B. Turkbey, and R. M. Summers, "DeepOrgan: Multi-level deep convolutional networks for automated pancreas segmentation," in *Proc. Int. Conf. Med. Image Comput. Comput.-Assist. Intervent.* Cham, Switzerland: Springer, 2015, pp. 556–564.

- [75] M. Knolle, G. Kaissis, F. Jungmann, S. Ziegelmayer, D. Sasse, M. Makowski, D. Rueckert, and R. Braren, "Efficient, high-performance semantic segmentation using multi-scale feature extraction," *PLoS ONE*, vol. 16, no. 8, Aug. 2021, Art. no. e0255397.
- [76] T. G. W. Boers, Y. Hu, E. Gibson, D. C. Barratt, E. Bonmati, J. Krdzalic, F. van der Heijden, J. J. Hermans, and H. J. Huisman, "Interactive 3D U-Net for the segmentation of the pancreas in computed tomography scans," *Phys. Med. Biol.*, vol. 65, no. 6, Mar. 2020, Art. no. 065002.



HIMALI GHORPADE received the B.E. degree in electronics and telecommunication engineering from Shivaji University, Kolhapur, Maharashtra, India, in 2015, and the M.E. degree in electronics and telecommunication engineering from Pune University, Maharashtra, in 2017. She is currently pursuing the Ph.D. degree in electronics and telecommunication engineering with the Symbiosis Institute of Technology, Pune, Maharashtra. She has been an Assistant Professor for three years

with the Department of Electronics and Telecommunication Engineering, Shrimati Kashibai Navale College of Engineering, Pune. Her current research interests include medical image processing, computer vision, machine learning, generative artificial intelligence, and video processing.



JAYANT JAGTAP (Senior Member, IEEE) received the B.E. degree in electronics and telecommunication engineering from Dr. Babasaheb Ambedkar Marathwada University, Aurangabad, Maharashtra, India, in 2008, the M.Tech. degree in electronics engineering from the Shri Guru Gobind Singhji Institute of Engineering and Technology, Nanded, Maharashtra, in 2011, and the Ph.D. degree in electronics and telecommunication engineering from Swami Ramanand

Tirth Marathwada University, Nanded, in 2017. He is currently an Assistant Professor with the Department of Electronics and Telecommunication Engineering, Symbiosis Institute of Technology, Pune, Maharashtra. His current research interests include digital image processing, pattern recognition, computer vision, medical image processing, and machine learning.



SHRUTI PATIL (Member, IEEE) received the M.Tech. degree in computer science and the Ph.D. degree in data privacy from Pune University. She has been an industry professional in the past, currently associated with the Symbiosis Institute of Technology as a Professor and SCAAI as a Research Associate, Pune, Maharashtra. She has three years of industry experience and ten years of academic experience. She has expertise in applying innovative technology solutions to real-

world problems. Her research interests include applied artificial intelligence, natural language processing, acoustic AI, adversarial machine learning, data privacy, digital twin applications, GANS, and multi-modal data analysis. She is also working in the application domains of healthcare, sentiment analysis, emotion detection, and machine simulation via which she is also guiding several bachelor's, master's, and Ph.D. students as a domain expert. She has published more than 30 research articles in reputed international conferences and Scopus/Web of Science indexed journals and books.



KETAN KOTECHA is currently an Administrator and a Teacher with the Symbiosis Centre for Applied Artificial Intelligence, Symbiosis International (Deemed University), Pune, India. His research interests include artificial intelligence, computer algorithms, machine learning, and deep learning. He has expertise and experience in cutting-edge research and projects in AI and deep learning for the last 25 years. He has published more than 200 articles widely in several excellent

peer-reviewed journals on various topics ranging from cutting-edge AI, education policies, teaching-learning practices, and AI for all. He was a recipient of the two SPARC projects worth INR 166 lakhs from the MHRD Government of India in AI in collaboration with Arizona State University, USA, and The University of Queensland, Australia. He was also a recipient of numerous prestigious awards, such as the Erasmus + Faculty Mobility Grant to Poland, the DUO-India Professors Fellowship for research in responsible AI in collaboration with Brunel University, U.K., the LEAP Grant with Cambridge University, U.K., the UKIERI Grant with Aston University, U.K., and a Grant from the Royal Academy of Engineering, U.K., under Newton Bhabha Fund. He has published three patents and delivered keynote speeches at various national and international forums, including with the Machine Intelligence Laboratory, USA, IIT Bombay, under the World Bank Project, the International Indian Science Festival organized by the Department of Science and Technology, the Government of India, and many more. He is an Associate Editor of the IEEE Access journal.

intelligence with Innopolis University, Russia, and the Yayasan Tun Ismail Mohamed Ali Professorial Chair of Artificial Intelligence with UCSI, Malaysia. He works in a multi-disciplinary environment, and he has authored/coauthored more than 1,400 research publications out of which there are more than 100 books covering various aspects of computer science. One of his books was translated to Japanese and a few other articles were translated to Russian and Chinese. He has more than 53,000 academic citations (H-index of more than 108 as per Google Scholar). He has given more than 150 plenary lectures and conference tutorials (in more than 20 countries). He was the Chair of the IEEE Systems Man and Cybernetics Society Technical Committee on Soft Computing (which has more than 200 members) (2008-2021). He served as a Distinguished Lecturer of the IEEE Computer Society representing Europe (2011-2013). He was the Editor-in-Chief of Engineering Applications of Artificial Intelligence (EAAI) (2016-2021). He serves/served on the editorial board of over 15 international journals indexed by Thomson ISI.



NATALLY HORVAT received the Ph.D. degree from the University of São Paulo. She is currently a Radiologist with the Memorial Sloan Kettering Cancer Center (MSKCC), New York, USA. Her research interests include abdominal radiology, oncologic body imaging, and the multimodality imaging of gastrointestinal cancers.



AJITH ABRAHAM (Senior Member, IEEE) received the B.Tech. degree in electrical and electronic engineering from the University of Calicut, in 1990, the Master of Science degree from Nanyang Technological University, Singapore, in 1998, and the Ph.D. degree in computer science from Monash University, Melbourne, Australia, in 2001. He is currently the Pro-Vice Chancellor of Bennett University, New Delhi, responsible for the University's Research and International Academic

Affairs. Prior to this, he was the Dean of the Faculty of Computing and Mathematical Sciences, FLAME University, Pune, and the founding Director of the Machine Intelligence Research Laboratories (MIR Laboratories), USA, a Not-for-Profit Scientific Network for Innovation and Research Excellence connecting industry and academia. He also held two international university professorial appointments, including a Professor in artificial



JAYASREE CHAKRABORTY (Senior Member, IEEE) received the B.E. degree in electronics and communication engineering from the National Institute of Technology, Durgapur, India, in 2005, the M.E. degree in electronics and telecommunication engineering from Jadavpur University, India, in 2008, and the Ph.D. degree in computer vision and machine learning from the Indian Institute of Technology (IIT) Kharagpur, Kharagpur, India, in 2013. Currently, she is an Assistant Attending

with the Memorial Sloan Kettering Cancer Center (MSKCC), Department of Surgery, New York, USA. Her research interests include machine learning, computer vision, medical image analysis, and image and video processing.